Author's response to reviews

Title: A cross-sectional survey of emergency and essential surgical care capacity among hospitals with high trauma burden in a Central African country

Authors:

Marquise E. Kouo-Ngamby (m.ngamby@gmail.com)
Fanny Dissak-Delon (fannynadia@gmail.com)
Isabelle Feldhaus (isabelle.feldhaus@ucsf.edu)
Catherine Juillard (catherine.juillard@ucsf.edu)
Martin E. Monono (ekekemononom@who.int)
Kent A. Stevens (ksteve14@jhmi.edu)

Version: 4 Date: 24 September 2015

Author's response to reviews: see over
Dear Editorial Team,

On behalf of our research team, I would like to submit our revised manuscript: A cross-sectional survey of emergency and essential surgical care capacity among hospitals with high trauma burden in a Central African country. Below, we have outlined responses and revisions according to the handling Editor’s requests:

1. Figure legends have been removed from figures and added to the end of the manuscript (page 22).
2. Figure 1 has been revised to appropriately attribute Google and its data providers in compliance with Google guidelines and Terms of Service with the addition of “Map data ©2015 Google” to the image.

Please let us know if there are any further revisions necessary. Thank you for your time and consideration of our manuscript submission.

Sincerely,

Marquise Kouo Ngamby, MD, MPH
Head of Maternal Health
Ministry of Public Health, Cameroon